Author's response to reviews

Title: Clinical trial: probiotic treatment of acute distal ulcerative colitis with rectally administered Escheria coli Nissle 1917 (EcN)

Authors:

Harald Matthes (hmatthes@havelhoehe.de)
Thomas Krummenerl (krummenerl@gastro-muenster.de)
Manfred Giensch (dr.med.m.giensch@t-online.de)
Corinna Wolff (corinna.wolff@ardeypharm.de)
Jürgen Schulze (corinna.wolff@ardeypharm.de)

Version: 11 Date: 6 April 2010

Author's response to reviews:

Dear editors,

we deleted the term "Aim" in the abstracts (manuscript as well as system).

Kind regards
harald Matthes